\documentclass{article}
%\usepackage[OT4,OT1,T1]{fontenc}
%\usepackage[latin1,cp1255]{inputenc}
\usepackage{array,amsfonts}
\makeatletter
%\usepackage{babel}
%\newcommand{\theauthor}[1]{\hfill\begin{otherlanguage}{english}{\it #1}\end{otherlanguage} }
%\font\scf=ptmr8r scaled 910
%\font\kukua=ptmr8r scaled 1370
%\font\kukub=ptmr8r scaled 1200
%\usepackage{ij}
%\usepackage{times}

\def\ {\thinspace}
\lineskiplimit=100pt 
\lineskip=6.9pt

\def\index\indexvolume#1\indexyear#2\endindextitle{
\rm\parindent=0pt%\overfullrule=0pt
\centerline{ISRAEL JOURNAL OF MATHEMATICS}
\vskip0.4cm
\centerline{TABLE OF CONTENTS, VOLUME #1\unskip, #2\unskip}
\vskip1.2cm
\frenchspacing}
\def\endindex{}


%\def\indexitem\indexauthor#1\indextitle#2\indexpage#3\endindexitem
%{\halign {\normalbaselines%
%\vbox{\hsize=11cm\hangindent=2em\hangafter=1 ##}%
%&\hbox to 1.3cm{\hfill ##}\cr%
%{\sc #1\unskip}\quad{\sl #2} & #3 \cr}\par}
%\def\indexauthor{}
%\def\endindexitem{}
%\def\indexpage{}
%\def\indextitle{}
\newcommand{\indexitem}[3]{\halign {\normalbaselines%
\vbox{\hsize=11cm\hangindent=2em\hangafter=1 ##}%
&\hbox to 1.3cm{\hfill ##}\cr%
{\sc #1\unskip}\quad{\sl #2} & #3 \cr}\par}

%\def\indexsee\indexauthor#1\partner#2\endindexitem
%{{\sc #1\unskip\enspace}{\sl See }{\sc #2\unskip}\par}

\newcommand{\indexsee}[2]{{\sc #1\unskip\enspace}{\sl See }{\sc #2\unskip}\par}

\makeatother

\begin{document}
\pagestyle{empty}

\index
\indexvolume 163
\indexyear 2008
\endindextitle


%1 3456
\indexitem{Dinh, T.-C. and N. Sibony}{Upper bound for topological entropy of a meromorphic correspondence}{29}
%2 3452
\indexitem{Finston, D. R. and S. Maubach}{The automorphism group of certain factorial threefolds and a cancellation problem}{369}
%3 3457
\indexitem{Foata, D. and G.-N. Han}{Signed words and permutations, IV: fixed and pixed points}{217}
%4 3466
\indexitem{Friedman, G.}{Cobordism of disk knots}{139}
%5 3457
\indexsee{Han, G.-N.}{D. Foata}
%6 3453
\indexitem{Hjorth, G.}{Non-treeability for product group actions}{383}
%7 3447
\indexitem{Hollander, S.}{A homotopy theory for stacks}{93}
%8 3463
\indexitem{Izumi, M., S. Neshveyev and R. Okayasu}{The ratio set of the harmonic measure of a random walk on a hyperbolic group}{285}
%9 3464
\indexitem{Jiao, Xiaoxiang}{Pseudo-holomorphic curves of constant curvature in complex Grassmannians}{45}
%10 3460
\indexitem{Kechris, A. S. and B. D. Miller}{Means on equivalence relations}{241}
%11 3462
\indexitem{Knutson, A., E. Miller and A. Yong}{Tableau complexes}{317}
%12 3452
\indexsee{Maubach, S.}{D. R. Finston}
%13 3446
\indexitem{Meyerovitch, T.}{Tail invariant measures of the Dyck shift}{61}
%14 3460
\indexsee{Miller, B. D.}{A. S. Kechris}
%15 63462
\indexsee{Miller, E.}{A. Knutson}
%16 3448
\indexitem{Moret\'o, A. and G. Navarro}{Groups with three real valued irreducible characters}{85}
%17 3448
\indexsee{Navarro, G.}{A. Moret\'o}
%18 3463
\indexsee{Neshveyev, S.}{M. Izumi}
%19 3463
\indexsee{Okayasu, R.}{M. Izumi}
%20 3467
\indexitem{Popov, S. I. and J.-M. Strelcyn}{On rational integrability of Euler equations on Lie algebra $so(4,\mathbb C)$}{263}
%21
\indexitem{Psaltis, P.}{A rigidity theorem for automorphism groups of trees}{345}
%22 3444
\indexitem{Randrianantoanina, N.}{Embeddings of non-commutative $L^p$-spaces into preduals of finite von Neumann algebras}{1}
%23 3438
\indexitem{\v Semrl, P.}{Endomorphisms of matrix semigroups over division rings}{125}
%24 3456
\indexsee{Sibony, N.}{T.-C. Dinh}
%25 3455
\indexitem{Sidi, A.}{A de Montessus type convergence study of a vector-valued rational interpolation procedure}{189}%26 3467
\indexsee{Strelcyn, J.-M.}{S. I. Popov}
%27 3462
\indexsee{Yong, A.}{A. Knutson}
%28 1111
\indexitem{}{\hspace{23pt}Editorial note}{411}
\end{document}